Author’s response to reviews

Title: The dating of thrombus organization in cases of pulmonary embolism: an autopsy study

Authors:

Celestino Sardu (drsarducele@gmail.com)
Gelsomina Mansueto (gelsomin.mansueto@unina.it)
Dario Costa (dario.costa@unicampania.it)
Emanuele Capasso (emanuele.capasso@unina.it)
Federica Varavallo (celestino.sardu@virgilio.it)
Giuseppina Brunitto (celestino.sardu@unicampania.it)
Rosanna Caserta (sarducelestino@libero.it)
Salvatore Esposito (drcelesardu@gmail.com)
Massimo Niola (massimo.niola@unina.it)
Celestino Sardu (drsraducele@gmail.com)
Raffaele Marfella (raffaele.marfella@unicampania.it)
Claudio Napoli (claudio.napoli@unicampania.it)
Mariano Paternoster (mariano.paternoster@unina.it)

Version: 4 Date: 27 Sep 2019

Author’s response to reviews:

Editor Comments:

1. Availability of Data and Materials

Please revise your current Availability of Data and Materials statement to reflect one of the formats indicated in our submission guidelines: https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials.
We thank the reviewer for his/her comment. As You suggested, we revised the current Availability of Data and Materials statement.

2. Authors’ Contributions

Please include a statement in the Authors’ contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

We thank the reviewer for his/her comment. As You suggested, we included a statement in the Authors’ contributions section to the effect that all authors have read and approved the manuscript.

3. Order of Authors in System

Please correct the order of authors in our system to match the title page.

We thank the reviewer for his/her comment. As You suggested, we corrected the order of authors in the system to match the title page.

4. Keywords

Please move the keywords after the abstract and before the main body of your manuscript.

We thank the reviewer for his/her comment. As You suggested, we moved the keywords after the abstract and before the main body of the manuscript.

5. List of Abbreviations

Please rename your “Abbreviations” to “List of Abbreviations”.

We thank the reviewer for his/her comment. As You suggested, we renamed the list of abbreviations to “List of Abbreviations”.
We thank the reviewer for his/her comment. As You suggested, we renamed the “Abbreviations” to “List of Abbreviations”.

6. Remove section

Please remove the section "Authors' Information Declaration section"

We thank the reviewer for his/her comment. As You suggested, we removed the section "Authors' Information Declaration section".

7. Remove files

Please remove the response to the reviewers comments and the cover letter files from the file inventory as it is no longer needed at this stage of the editorial process.

We thank the reviewer for his/her comment. As You suggested, we removed the response to the reviewers comments and the cover letter files from the file inventory as it is no longer needed at this stage of the editorial process.

8. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

We thank the reviewer for his/her comment. As You suggested, we uploaded a clean version that did not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.